

Contents lists available at ScienceDirect

International Journal of Production Economics

journal homepage: www.elsevier.com/locate/ijpe



Productivity with endogenous FDI spillovers: A novel estimation approach

Ioannis Bournakis^{a, *}, Mike Tsionas^{b, c}

^a SKEMA Business School, Université Côte d'Azur, France

^b Montpellier Business School, France

^c Lancaster University Management School, UK

ARTICLE INFO

Keywords: TFP Productivity Endogeneity FDI spillovers Reallocation effects MNEs

JEL classification: F2 O4

ABSTRACT

The paper develops a new methodological framework for evaluating the role of FDI in the domestic economy. We firstly propose a measurement of Total Factor Productivity (TFP), which accounts for endogenous FDI knowledge spillovers. Our estimation allows knowledge spillovers to co-evolve with output and inputs selection, ensuring that the estimated production inputs are consistent whilst correctly identifying performance improvements. After deriving unbiased TFP measures at the firm level, we aggregate them at the industry level to search for reallocation effects driven by the FDI presence. Our methodology distinguishes between within-firm gains and industry reallocation effects from FDI. We apply our novel framework to a sample of 7699 manufacturing firms from six EU countries. The main findings are: (i) endogenous FDI spillovers correct for the omitted variable bias in the estimation of production inputs; (ii) inter-industry spillovers are important sources of TFP gains; (iii) the realization of gains from FDI knowledge spillovers are dependent on the absorptive capacity of the firm (iv) higher levels of FDI presence in the local economy can contribute to aggregate TFP increases as much as 33%. The paper offers the basis for considering new policy perspectives on FDI incentives and suggests new approaches for modeling the mechanisms through which domestic firms can experience productivity gains from their interaction with foreign counterparts in a globalized business environment.

1. Introduction

A highly contentious issue in production economics is the estimation and evolution of productivity. The literature documents substantial Total Factor Productivity (TFP) differentials even within narrowly defined industries (Decker et al., 2016; Hsieh and Klenow, 2009; Syverson, 2004), which always raises the question of what are the drivers behind these differentials. Potential candidates for explaining these disparities include mechanisms of globalization (e.g. changes in the trade regime, FDI activity), innovation (e.g. R&D) and alternations in the fiscal environment (Bournakis and Mallick, 2021). The interest in understanding the sources of productivity differences within and across countries (De Loecker and Goldberg, 2014) is related to the direct link between productivity improvements and welfare gains.¹ As firms increase their productivity, these improvements are translated into aggregate gains (industry and national) that enable a country to achieve higher standards of living. Despite the importance of the topic, some of the determinants of productivity are not appropriately elaborated in the firm's production process, which causes inaccurate and misleading inferences. One of the main contributions of the present paper is to object to the methodological approach currently employed in evaluating the impact of FDI on the performance of domestic firms.

So far, the effect of FDI-related spillovers on the productivity of domestic firms encompasses regressions of TFP on indices of foreign knowledge spillovers (Bournakis, 2021; Iršová and Havránek, 2013; Havránek and Iršová, 2011). This analytical framework provides reasonable point estimates for the economic importance of spillovers, but it does so in an ex-post fashion without treating spillovers as integral components of the production function (Javorcik, 2004; Lu et al., 2017; Newman et al., 2015). Assuming that FDI spillovers evolve independently from productivity, as the current literature does, is a serious limitation, which does not allow for external knowledge to interact endogenously with the selection of capital and labour in the production process. Hence, results that presume spillovers are exogenous to output

* Corresponding author.

https://doi.org/10.1016/j.ijpe.2022.108546

Received 17 January 2022; Received in revised form 4 June 2022; Accepted 9 June 2022 0925-5273/© 20XX

E-mail address: ioannis.bournakis@skema.edu (I. Bournakis).

¹ A google scholar search with keywords "globalization" and "productivity growth" returns almost 253,000 papers, which shows the substantial interest gathered in the topic from academics and policymakers.

can mislead us about the true impact of FDI on domestic firms' performance.²

In theory, knowledge spillovers are expected to generate learning gains that boost the TFP of domestic firms.³ TFP gains of this type are within-firm improvements. Additionally, FDI enhances reallocation effects at the aggregate level. Embracing foreign knowledge helps domestic firms become more productive, increasing market share and outperforming less productive rivals. FDI also fosters competition within the industry and, in conjunction with the gains from learning within firms, enhances sectoral productivity. Although within-firm productivity gains and industry reallocation gains represent significant welfare channels improvements, the regression analysis that has been used for evaluating FDI spillovers does not separate the two. This severe limitation defines the departure point of our paper that assumes endogenous evolution of FDI spillovers (within-firm effects). It then searches for aggregate gains that emerge from the reallocation dynamics within the industries (reallocation effects). Our methodology addresses two significant points; first, spillovers are allowed to co-evolve endogenously with firm productivity in the production function. We suggest an estimation procedure that accounts for the effect of external knowledge spillovers in the derivation of input estimates in the production function. Second, we search for market share reallocation effects driven by the FDI presence in the industry after domestic firms have improved productivity from the absorption of FDI knowledge spillovers. By distinguishing between within-firm and across-firm effects, our estimation approach moves the literature beyond the current state of simple regressions of domestic TFP on FDI to a more systematic evaluation of FDI in the domestic economy.

In the first part of the paper, we depart from the estimation methodology of De Loecker and Warzynski (2012) to endogenize FDI knowledge spillovers in the output process of the firm. Within this behavioral framework,⁴ we estimate the parameters of the production function controlling for unobserved productivity using observables. Specifically, we use the demand function of intermediate input materials to proxy for unobserved productivity shocks ala Levinsohn and Petrin (2003).⁵ A two-stage method is adopted. In the first stage, the demand of materials is used as a control function to obtain the expected output (De Loecker and Warzynski, 2012; Ackerberg et al., 2015). The second stage recovers estimates of all inputs and TFP. We examine alternative specifications of how FDI knowledge spillovers govern productivity in the second stage of the estimation procedure. In addition to spillovers, intangible capital is another important productivity shifter in the second stage. Biased estimate coefficients of the production function result from neglecting activities that are expected to be correlated with the use of inputs, thus productivity. Biased estimated coefficients of the production function are caused by neglecting firm activities that are expected to be correlated with the use of inputs, thus productivity. In this respect, coefficients of the non-dynamic inputs are upward biased (labor and intermediate materials), while the quasi-fixed input of capital is downward biased (Olley and Pakes, 1996). Therefore, our paper also serves as a guide to a modified law of motion model of productivity (De Loecker, 2013; De Loecker and Goldberg, 2014) that addresses the omitted variable bias caused by excluding productivity-enhancing factors correlated with the use of inputs.

In the second part of the paper, we aggregate firm TFP at the twodigit industry level to search for reallocation effects. We are investigating whether industries that attract higher levels of FDI achieve a higher level of TFP due to reallocation gains. The latter implies that the absorption of knowledge spillovers makes domestic firms obtain higher market shares, promoting industry TFP. This approach is unique in the literature to the best of our knowledge. It studies the effect of FDI at the sectoral level after aggregating micro-firm level TFP that already internalizes knowledge spillovers.

The paper's contributions are fourfold: (a) we separate the sources of gains from FDI. Prior literature mixes knowledge gains from the advanced know-how of MNEs with the reallocation dynamics emerging from the exposure of the domestic market to higher levels of competition. Instead, we distinguish between the two, which analyses more systematically the role of FDI in the domestic economy; (b) we endogenize FDI knowledge spillovers within a semi-parametric estimation of firmlevel TFP, reducing the omitted variable bias, whilst ensuring that sources of productivity improvement are adequately identified; (c) we inform the law of motion of productivity with other factors of internal knowledge capital (i.e.intangibles) that also interact endogenously with FDI spillovers and inputs selection; (d) since we identify different sources of FDI gains, our findings provide the basis for a more tuned policy design, which can potentially target specific aspects of FDI to maximize the benefits for local firms.

We apply our methodological framework to a sample of 7699 manufacturing firms from six EU countries using the European Firms in the Global Economy (EFIGE) data set. Findings indicate that incorporating spillovers into the evolution of productivity leads to higher TFP estimates; however, not all spillovers positively impact domestic TFP. For domestic firms to increase productivity, they must form inter-industry partnerships with multinational organizations (MNEs). Productivity gains from knowledge spillovers depend on domestic firms' absorptive capacity. The analysis reveals that higher levels of FDI presence in the local economy can contribute to reallocation gains of as much as 33%. Aggregate (sectoral) productivity gains from reallocation dynamics appear to generate a one-off effect without strong time persistence. The paper is structured as follows: section 2 outlines the links between spillovers and productivity, section 3 describes the estimation methodology, section 4 describes the data and the results obtained from the estimation of the productivity parameter under different frameworks, section 5 shows TFP results with endogenous FDI spillovers, section 6 estimates the reallocation effects, section 6 conducts robustness analysis and section 7 concludes the paper.

2. Spillovers and firm productivity

Measurement of productivity at the micro-level is primarily concerned with avoiding simultaneity bias resulting from unobserved productivity shocks and input selection. The econometrician may not be aware of contemporaneous productivity shocks known to the firm manager. In the presence of simultaneity bias, standard OLS produces estimates of the production input that are upward biased. The previous literature uses the demand of investment Olley and Pakes (1996) and materials Levinsohn and Petrin (2003) to approximate unobserved productivity shocks. These estimation techniques rely on two assumptions about the nature and the timing of inputs selection. Concerning nature, inputs are distinguished into dynamic and non-dynamic. Accordingly, the cost of a dynamic input affects the future profit of the firm (capital), while as regards timing, some inputs can be chosen within the same period (labour) and inputs that are determined in the year before (capital). On the basis of these assumptions, Olley and Pakes (1996) and Levinsohn and Petrin (2003) derive estimates of TFP using a two-stage procedure. The first stage generates expected output, a coefficient of labour (non-dynamic input) and an estimate of the usual statistical noise of the residuals. The second stage specifies a first-order exogenous

 $^{^{2}}$ Crespo and Fontoura (2007) provide an extensive conceptual discussion about the role of FDI in the domestic economy.

³ Gains from FDI include the diffusion of advanced technologies and better management and organizational practices from MNEs, as well as the possibility of hiring personnel trained by MNEs (Meyer and Sinani, 2009).

⁴ The firm solves a common cost minimization problem to obtain the optimal input demand and output elasticity of a variable input without adjustment costs.

⁵ Using the demand for investment to control for unobserved productivity shocks, Olley and Pakes (1996) offers the first semi-parametric estimation of firm-level TFP.

Markov process of productivity and recovers an estimate of capital (dynamic input).⁶ The framework we propose relaxes this restriction by assuming that productivity evolves endogenously with knowledge spillovers.

Two are the core questions regarding the implementation of our framework: how knowledge spillovers occur and subsequently how they govern firm productivity. In an increasingly globalized environment, on the one hand, firms are exposed to severe competition, while on the other hand, they learn from interaction with international counterparts. This internationalization process makes Multinational Enterprises (MNEs) and domestic firms develop partnerships mainly in inputs transactions. These are either industrial linkages of domestic firms with MNEs in upstream industries that sell inputs or linkages with MNEs in downstream industries that buy inputs from domestic firms. The contracts, which formalize these inter-industry transactions, do not constitute spillovers per se. However, the commodities that domestic firms purchase from MNEs embody the producer's know-how, while those commodities that MNEs order from domestic firms should satisfy the buyer's technological requirements. These vertical (inter-industry) linkages encompass tacit knowledge that contributes to the domestic firm's internal expertise, thus potentially increasing performance and efficiency.

Moreover, the presence of MNEs within the same industry exposes the domestic market to higher levels of competition. As a response to competition pressure oriented from MNEs, domestic firms reorganize operations and production to increase efficiency and better allocate the existing resources. Domestic firms replace older managers, readjust scale size, and (or) reorganize the production chain more costeffectively. FDI, either in the form of inter-industry knowledge spillovers or intra-industry competition effects, creates an array of opportunities and challenges that can affect the productivity of domestic firms. It is not guaranteed that FDI will positively impact the host economy. The presence of MNEs in a domestic market is likely to crowd out domestic firms (i.e. market stealing effect), causing productivity losses. Most commonly, MNEs create agglomeration effects that stimulate the diffusion of knowledge in the local market (Aitken and Harrison, 1999; Orlando, 2004). Although the manager of the domestic firm observes these effects, they remain unobserved to the researcher. From a modeling point of view, if we fail to account for knowledge spillovers in the evolution of physical productivity, the estimated coefficients of the production function remain highly biased.

Bias emerges not only if we neglect spillovers but also in other forms of knowledge capital, such as intangible assets. To reduce the bias caused by omitted variables, we consider the increasing role of intangible assets in firm performance, which represent a firm's organizational capital.⁷ Recent literature (Crass and Peters, 2014) has demonstrated that intangible capital as a measure of organizational practices influences innovation activities and has a positive impact on productivity (Bloom et al., 2010, 2012). Marrocu et al. (2012) show that some components of intangible assets tend to be as productive as tangible capital.⁸ Firm management and organizational practices are complementary to spillovers in our context because they determine a firm's ability to assimilate external knowledge. The combination of the two determines absorptive capacity, an important driver of productivity (Escribano et al., 2009; Aldieri et al., 2018).

3. Estimation methodology

In our structural framework, production input coefficients correspond to cost input shares consistent with the existence of input adjustment costs. This assumption follows the formulation of Ackerberg et al. (2015), while it differs from Olley and Pakes (1996) and Levinsohn and Petrin (2003) that consider labour as a free input without cost adjustments. Within an EU context with strong employment protection legislation (EPL) in place, it becomes more realistic to assume that firm decisions about labour hiring maintain a more persistent impact on future profits. The two underlying assumptions of our structural framework are that producers are cost-minimizers, and technological change is Hicks-neutral. We specify a generic form of a log value-added production function as follows:

$$y_{ict} = f(l_{ict}, k_{ict}; \beta) + v_{ict}, \tag{1}$$

where *l* stands for labour, *k* stands for the capital stock and β are parameters of interest to be estimated. The usual notation applies with *i* and *t* to index firms and years, respectively. Although our panel includes a country index *c*, it is suppressed hereafter to improve readability. Residual v_{it} can be further decomposed into:

$$v_{it} = \omega_{it} + u_{it},\tag{2}$$

where ω_{it} is an unobserved factor that affects a firm *i*'s output (productivity), and u_{it} is the standard statistical noise that captures random errors. In industrial organization and trade literature, parameter ω_{it} is driven by exporting (De Loecker, 2013), changes in trade policy and within-firm efforts to reduce X-inefficiencies (De Loecker and Goldberg, 2014). In the present context, we shed light on spillovers and how they interact with characteristics within firms, such as intangible assets as the major sources of spillovers ω_{it} . In a reduced static form representation, ω_{it} is essentially a function:

$$\omega_{it} = h(spill_{it}, X_{it}, spill_{jt} \times X_{it}), \tag{3}$$

where *spill_{jt}* stands for industry-wise spillovers and X_{it} represents firm specific characteristics. For the purposes of the estimation framework, ω_{it} is assumed to follow a dynamic pattern (the law of motion) that captures the high degree of persistence in productivity.

The functional form of (1) is the translog that allows for greater flexibility between inputs. We also nest the baseline translog with a Cobb-Douglass production function for comparability (see results in Appendix B). To recover estimates for β that are preceding the derivation of TFP (ω_{it}), we first need to address the simultaneity bias between inputs choice and unobserved ω_{it} . Following Levinsohn and Petrin (2003) and Ackerberg et al. (2015), we rely on the use of materials as a proxy for unobserved productivity shocks. We specify the demand of materials as follows:

$$m_{it} = m(\omega_{it}, k_{it}), \tag{4}$$

with the inverse function of *m* to give us an expression of ω_{it} as follows: $\omega_{it} = \Omega(m_{it}, k_{it})$. If $(\partial m_{it}/\partial \omega_{it}) > 0$, then function $\Omega(.)$ is a good approximation of ω_{it} in the sense that a positive productivity shock means higher consumption of intermediate material inputs. Melitz and Levinsohn (2006) and Aw et al. (2011) have shown that as long as more productive firms charge only ordinarily higher mark-ups than non-productive firms, then the monotonicity condition of materials in ω_{it} holds with productivity shocks to reduce marginal costs raising the demand of materials and output at any given level of demand. While input estimates β and ω_{it} are recovered in the second stage, first stage is used to run:

⁶ In the Olley and Pakes (1996) jargon, the dynamic inputs of capital and age are called state variables, while labour is called a free variable.

⁷ Appendix A shows how the ratio of intangible to tangible assets evolves over the period for each country. Apart from Spain, the graphs depict an upward trend in the relative use of intangibles within the firm.

⁸ Intangible assets include R&D expenditure, patents, copyrights, software, employee training, trademarks, and other similar items. More importantly, Efige has extracted these data from Amadeus-Bureau Van Dijk (BDV), which has previously harmonized expenses on intangible assets across countries following international accounting rules. Therefore, our variable of intangible assets has been capitalized, and it is fully comparable across countries.

$$y_{it} = \varphi(l_{it}, k_{it}, m_{it}) + u_{it},$$
(5)

where we obtain predicted output $\hat{\varphi}$ and an estimate of the statistical noise \hat{u} from the following translog specification augmented with the elements of the inverse materials function:

$$\varphi_{it} = \beta_l l_{it} + \beta_k k_{it} + \beta_{ll} l_{it}^2 + \beta_{kk} k_{it}^2 + \beta_{lk} l_{it} k_{it} + \Omega(m_{it}, k_{it}).$$
(6)

The control function $\Omega(m_{it}, k_{it})$ accounts for productivity in the production function estimation in the first stage.

Second stage specifies an endogenous productivity process that allows for last year's external foreign knowledge (spillovers) to interact with in-house intangible assets in governing ω_{it} . The first-order Markov process (AR1) is specified as:

$$\omega_{it} = h(\omega_{it-1}, \mathbf{z}_{it-1}) + \psi_{it}, \tag{7}$$

where ω_{it} is estimated by:

$$\widehat{\omega}_{it}(\beta) = \widehat{\varphi}_{it} - \beta_l l_{it} - \beta_k k_{it} - \beta_{ll} l_{it}^2 - \beta_{kk} k_{it}^2 - \beta_{lk} l k_{it}.$$
(8)

We collect in \mathbf{z}_{it} all the additional drivers of ω_{it} : $\sum_{k=3}^{k=3} spill_{it-1}$ is the sum of the industrial knowledge spillovers with the summation to run over the three FDI spillover indices, $\kappa = [horizontal, backward, forward];$ int_{it-1} is a measure of firm i's specific organizational capital represented by intangible assets; $\sum_{k=3}^{\kappa=3} spill_{it-1} \times int_{it-1}$ is a term of a firm *i*'s absorptive capacity.⁹ As mentioned the interaction term $\sum_{k=3}^{k=3} spill_{ii-1} \times int_{ii-1}$ measures the capability of the firm to assimilate external discoveries and capitalize on productivity from foreign knowledge. These capabilities depend on the organizational profile of the firm, and are referred to in the literature as "absorptive capacity" (Aldieri et al., 2018; Griffith et al., 2003; Blalock and Gertler, 2009). Term ψ_{it} represents an i.i.d shocks common across firms and years. Equation (7) regresses nonparametrically the predicted output φ_{it} from the first stage on past values of ω_{it} , spillovers, intangible assets and absorptive capacity. This formulation relaxes the standard assumption of exogenous ω_{it} offering a more realistic set-up within which knowledge spillovers feedback directly in firm *i*'s productivity.

Equation (7) represents the second stage of our estimation algorithm and retrieves estimates for input parameters β and productivity ω_{it} Various alternative techniques exist for estimating (7). Originally Olley and Pakes (1996) suggest a polynomial or a Kernel function. Rizov and Walsh (2009) use a fourth-order polynomial, which yields very similar results to the non-parametric Kernel function. In the present methodology, we apply GMM with cluster robust standard errors at the firm level following the original estimator used in De Loecker and Warzynski (2012). A crucial aspect of the GMM estimator is identifying the momentary conditions. We allow lagged labour and its squared term to be an instrument for current labour. The exogeneity condition for such assumption, $E[l_{it-1}v_{it}] = 0$ implies that only contemporaneous labour reacts to productivity shocks. Lagged labour values are valid instruments of current labour if there is no correlation between the two over time. The second instrument considered is the current value of the capital stock, the state variable, and its squared term. The momentary conditions are summarized below:

$$E [l_{it-1} \psi_{it}] = 0,$$

$$E [k_{it} \psi_{it}] = 0,$$

$$E [k_{it}^2 \psi_{it}] = 0,$$

$$E [k_{it}^2 \psi_{it}] = 0,$$

$$E [(k_{it} l_{it-1}^2) \psi_{it}] = 0,$$

(9)

 $E\left[\left(k_{it}^{2}l_{it-1}^{2}\right)\psi_{it}\right] = 0.$

Once, we have retrieved GMM estimates for input parameters $\beta = \{\beta_l, \beta_k, \beta_{ll}, \beta_{kk}, \beta_{lk}\}$, we can obtain ω_{it} by:

$$\omega_{ii} = \widehat{\varphi}_{ii} - \widehat{\beta}_i l_{ii} - \widehat{\beta}_k k_{ii} - \widehat{\beta}_{li} l_{ii}^2 - \widehat{\beta}_{kk} k_{ii}^2 - \widehat{\beta}_{lik} l_{ii} k_{ii}.$$
(10)

We calculate ω_{it} under different formulations of the law of motion (7) starting from a parsimonious specification without spillovers and then we augment gradually with additional determinants. The empirical results from these calculations are discussed in section 5.2. To verify specifically the impact of spillovers on ω_{it} (productivity), we show results separately from the parametric estimation of (7):

$$\omega_{it}(\beta_k,\beta_l) = \theta_1 \omega_{it-1}(\beta_k,\beta_l) + \mathbf{z}'_{it-1}\Theta_z + \psi_{it}.$$
(11)

Equation (11) is an integral part of the second stage in our estimation framework, which also derives the coefficients of capital and labour. We consider the same GMM momentary conditions described in (8) with vector Θ to include parameters to be estimated of the variables in z_{it-1} . For simplicity, we show results in Table 3 only from the four variants of (11), which include interchangeably an aggregate index of spillovers, intangible assets and the interaction term of absorptive capacity.

4. Data and measurement issues

4.1. The Efige data set

Bruegel provides the Efige data, a Belgian non-profit international organization that gathered together a survey and balance sheet information from 7699 manufacturing firms (with 10 employees and above) from France, Germany, Hungary, Italy, Spain, UK over the period 2001–2014. Estimating the production function in two stages as per the specifications (1) to (8), we use value-added (y), defined as the operating revenue (OPRE) minus costs of materials (MATE); capital stock (k) measured by the book value of fixed assets (FIAS); labour (l) measured by the number of employees (EMPL) and wages measured by staff remuneration (STAF). We remove observations from the sample with missing and negative values for FIAS, OPRE, and MATE as they are undefinable for the log specifications (5), (7) and (9). We deflate nominal euro values of OPRE, MATE and FIAS using a 2-digit NACE industry production price index (2005 = 100) from Eurostat. Appendix C shows summary statistics of the deflated variables used to estimate the production function. Furthermore, the Efige dataset provides information about the ownership status of the firm and its export activities. Specifically, we define a firm as a foreign MNE if the first shareholder is of foreign nationality and owns at least 10% of the capital shares (IMF, 2009). We utilize the following items from the Efige survey regarding export status: (a) which percentage of your 2008 annual turnover did the export activities represent?, and (b) has the firm exported before 2008? Based on the two items, we define *export* = 1 if *foreign sales* > 0in 2008 and firm i always exports before 2008; 0 otherwise.¹⁰ This definition includes only firms that are established exporters with exports throughout the entire period of the sample. Appendix D shows the total

⁹ The current data Efige availability imposes the use of intangible assets in an aggregate form instead of using separate components.

¹⁰ Question coded D4 in the Efige questionnaire asks: Which percentage of your 2008 annual turnover did the export activities represent? and question coded D5 asks: Before 2008, has the firm exported any of its products? The available answers to these questions are: (i) always, (ii) sometimes, and (iii) never.

4.2. Spillovers

FDI is expected to generate spillovers through strengthening competition in the domestic market, unintended transfer of sophisticated technology from MNEs, dissemination of advanced organizational and managerial practices, and the establishment of better distributional networks. We identify three channels through which domestic firms can benefit from these spillovers: (a) intra-industry spillovers (*horizontal*) from MNEs in the same industry, (b) inter-industry spillovers (*forward*) through the purchase of inputs from MNEs in upstream industries, and (c) inter-industry spillovers (*backward*) through the sale of inputs to MNEs in downstream industries. The effect from intra-industry linkages (horizontal spillovers) is not favorable by default as it is possible to capture both gains from higher competition but also crowing out effects on domestic firms within the same 3-digit level industry:

$$Horizontal_{jct} = \frac{\sum_{F \in j} S_{Fjct}}{\sum_{i \in j} S_{ijct}},$$
(12)

where *S* is the sales revenue of MNEs (*F*) to total sales in industry *j* (3-digit NACE. Rev2) at year *t* (country index c is again suppressed for readability).

Forward and backward spillovers represent vertical connections between MNEs and domestic firms. Forward spillovers are derived from MNEs inputs suppliers with:

$$Forward_{jct} = \sum_{i \neq h}^{J-1} \gamma_{jh} Horizontal_{hct},$$
(13)

where γ_{jh} is the coefficient of the Leontief input-output inverse matrix (OECD, 2012b) that represents the amount of intermediate output purchased from upstream industry *h* in order to produce one unit of output in the downstream industry *j* at year *t*.¹¹ *Horizontal* measures horizontal spillovers in the upstream industry *h*. Analogously, backward spillovers are derived from MNEs input buyers:

$$Backward_{jct} = \sum_{i \neq w}^{J-1} \gamma_{jw} Horizontal_{wct}, \tag{14}$$

where γ_{jw} is now the coefficient that represents the amount of intermediate output purchased from upstream industry *j* in order to produce one unit of output in downstream industry *w*. Coefficients γ_{jh} and γ_{jw} are time invariant parameters that captures transactions at the 2-digit industry level (OECD, 2012a). In our empirical implementation, we use summation of the three indices:

$$spill_{jt} = Horizontal_{jt} + Forward_{jt} + Backward_{jt}.$$
 (15)

We also separate between horizontal and vertical. The latter is the sum of forward and backward spillovers. By disentangling the two, we check whether horizontal spillovers induce adverse effects through "market stealing" that prevents domestic firms from exploiting economies of scale. Not achieving a certain size threshold can be detrimental and potentially a source of slack for the performance of domestic firms. Nonetheless, vertical spillovers are expected to only positively affect productivity as they are derived from economic transactions in the form of purchase and sale of inputs between technologically superior MNEs and laggard domestic firms.

5. Empirical results

After considering various formulations of vector z_{it} and different assumptions about the evolution of ω_{it} in equation (7), we estimate nine specifications. We start with a benchmark specification S1 of a valueadded translog production function that allows for a first-order Markovprocess of ω_{it+1} without controlling for any source of spillovers in the second stage. S1 is identical to the estimation framework suggested by De Loecker and Warzynski (2012). **S2** derives ω_{it+1} from a translog production function that assumes industry homogeneity in the production input parameters and also allows for $\sum_{k=3}^{k=3} spill_{it}$ to enter the evolution of ω_{it+1} in the second stage. S3: S2 with industry heterogeneity in the production input parameters. S4: S2 with only vertical_{it} to be included in the Markov process in the evolution of ω_{it+1} in the second stage. S5: S2 with only *horizontal*_{it} to be included in the evolution of ω_{it+1} . S6: S2 plus investment in intangibles (intit) and absorptive capacity $(int_{it} \times \sum^{\kappa=3} spill_{it})$ entering z_{it} in the Markov-process in (7). **S7:S2** but with a second order Markov-process in the evolution of ω_{it+1} in the second stage. S8 is a third order translog production function in inputs with $\sum_{i=1}^{\kappa=3} spill_{it}$ entering z_{it} in the evolution of ω_{it+1} . To capture the notion that exporters produce under a different technology (López, 2009; Farinas and Martín-Marcos, 2007), we run S9, a value-added translog production function with an export dummy as an additional input and $\sum^{\kappa=3} spill_{it}$ in the evolution of ω_{it+1} . Appendix B shows estimates of a Cobb-Douglass value-added production function with industry heterogeneity and two variant specifications for the evolution of ω_{it} in the second stage; one without spillovers and one with an aggregate index of spillovers, $\sum_{k=3}^{\kappa=3} spills_{it}$. Finally, Appendix B shows also results from a translog production function with a dummy for domestic multinationals as an additional input. Table 2 summarizes the description of each specification and shows the sample means of TFP. We provide a further discussion of these results in the next section.

5.1. Estimates of the production function

The main objective of the estimation technique is to include productivity factors that are potentially correlated with the selection of inputs. We mitigate the omitted variable bias from assuming an exogenous evolution of productivity by augmenting (7) with spillovers and intangible assets (organizational capital). Our approach ensures that the estimated production inputs are consistent while performance improvements of the firm are appropriately identified. Without considering, for example, knowledge spillovers in the law of motion, equation (7), some variation in output can be mistakenly attributed to inputs, while the true source of improvement is learning from FDI spillovers. We consider S3 and S6 to be our baseline specifications as they control for the core elements of the analysis; these are spillovers, organizational capital, and absorptive capacity. Excluding these factors, from the evolution of ω_{it} causes potentially biased input coefficients. Table 2 presents the coefficient estimates of the translog production function for five specifications, including S1 for comparability despite not accounting for spillovers.

Comparing input coefficients between S1 and S3 in Table 2, there are considerable differences concerning the labour input. S3 with spillovers in the process of ω_{it} generates a labour coefficient smaller by 8.5% (0.87) relative to S1 (0.95). As we include additional productivity enhancers in S4 and S6, the labour coefficient decreases further. The labour coefficient reaches the value of 0.61 in S6 when the term of absorptive capacity is present in the process of ω_{it} . As expected the coefficient of capital increases to 0.37 after the inclusion of spillovers and absorptive capacity in the law of motion in S6. Nonetheless, the downward bias in the capital coefficient from neglecting spillovers is some-

¹¹ Assume that total output in the economy is described by the following expression: $\mathbf{y} = \widetilde{\Gamma} \mathbf{x}$, where $\widetilde{\Gamma} \mathbf{x}$ is the demand for intermediate products that should be consumed for the production of the final output. The solution of this linear programming problem in an economy with J sectors can be written as: $y = (I_J - \widetilde{\Gamma})^{-1}$, with I_J to be the $J \times J$ identity matrix and $(I_J - \widetilde{\Gamma})^{-1}$ to be the Leontief inverse matrix. Each element of this matrix is a non-negative number smaller or equal than one.

I. Bournakis and M. Tsionas

Table 1

Estimated TFP (ω) under different framewor	ks.
---	-----

Spec	Description	Mear
S1	Translog with endogenous ω_{it+1} in the 2nd stage without spillovers	3.11
S2	Translog with industry homogeneity and $\sum_{k=3}^{\kappa=3} spill_{jl}$ in the evolution of	4.25
	ω_{it+1}	
S 3	Translog with industry heterogeneity and $\sum_{k=3}^{k=3} spill_{jl}$ in the evolution of	3.50
	ω_{it+1}	
S4	S3 with only <i>vertical</i> _{jt} in the evolution of ω_{it+1}	3.82
S5	S3 with only $horizontal_{jt}$ in the evolution of ω_{it+1}	3.36
S6	S3 + (<i>int_{it}</i>) + (<i>int_{it}</i> × $\sum_{k=3}^{k=3}$ spill _{jt}) in the evolution of ω_{it+1}	4.31
S7	Translog with 2nd order Markov precess in ω_{it+1} that allows for $\sum^{\kappa=3} spill_{j_l}$	5.74
S8	Third order Translog in inputs that allows for $\sum_{k=3}^{k=3} spill_{j_l}$ in ω_{it+1}	1.64
S9	Translog with an export dummy as production input and $\sum_{j=1}^{\kappa=3} spill_{jt}$ in the evolution of ω_{it+1}	4.06

Table 2

Production function estimates.

	S1	S3	S4	S 6	S7
β_l	0.95	0.87	0.81	0.61	0.87
	(0.14)	(0.09)	(0.15)	(0.22)	(0.24)
β_k	0.32	0.37	0.34	0.37	0.35
	(0.06)	(0.05)	(0.08)	(0.12)	(0.06)
β_{ll}	0.07	0.07	0.10	0.13	0.07
	(0.15)	(0.12)	(0.02)	(0.07)	(0.05)
β_{kk}	0.02	0.02	0.02	-0.01	0.03
	(0.04)	(0.01)	(0.2)	(0.15)	(0.01)
β_{kl}	-0.11	-0.11	-0.12	-0.03	-0.10
	(0.05)	(0.05)	(0.39)	(0.08)	(0.03)

Notes: Cluster robust standard errors at the firm level are shown in parentheses.Coefficients in bold indicate significance at 5% and above. Refer to Table 1 for a full description of each specification (S).

how more negligible. Overall, Table 2 indicates that the augmentation of ω_{it} in S6 with spillovers tends to decrease the overestimation of labour relative to S1, which assumes productivity to evolve independently from FDI spillovers. To a lesser extent, S6 also corrects for underestimating the elasticity of capital. The upward bias in the coefficient of labour is 55% (i.e. the percentage difference in the $\hat{\beta}_l$ between S1 and S6), and it is mainly caused by assuming that FDI spillovers are exogenous. Intuitively, this finding implies that significant learning externalities have been attributed to labor, but are primarily the result of MNEs'presence in the domestic market, the internal organization capital of the firm, and the interdependence between both.

5.2. TFP measures under different estimation frameworks

Turning to the TFP results from different specifications as outlined in Table 1, the first remark is the existence of substantial industry heterogeneity as shown by the differences in the values of ω between S2 and S3. When comparing S1 and S3, a higher value of ω_{it} emerges in S3, which reinforces our view that productivity evolves endogenously. Separating the aggregate spillovers index into vertical and horizontal components reveals significant differences in ω_{it} . S4 includes only vertical spillovers with a sample average value of TFP, 3.82, 13% higher in comparison to S5 which includes only horizontal spillovers.

S3 generates a higher value of ω_{it} than S5 suggesting that FDIrelated spillovers can be counter-productive as they can induce "market stealing" effects that undermine the capacity of domestic firms to benefit from economies of scale within the same narrowly defined industry. S4 highlights that vertical spillovers (inter-industry linkages) are the main source of productivity gains from FDI. Partnerships between domestic firms and MNEs, whether through the supply of inputs or the purchase of inputs, are among the most important channels of knowledge exchange.

Absorptive capacity (S6) $int_{it} \times spil_{jt}$, has a positive impact on productivity and increases the average value of ω_{it} to 4.31. Following the previous discussion, the in-house organizational efforts of the firm to improve innovation capabilities generate substantial productivity gains that can be as close as to 19% relative to specification S3. A second-order process in the evolution of ω_{it} (S7) produces the highest value of ω_{it} among all the alternative frameworks in Table 1. This is due to the high degree of persistence in the evolution of productivity, which further signifies the importance of knowledge accumulation partly driven by knowledge spillovers. A third-order translog in inputs with the augmentation of spillovers in the second stage (S8) produces the lowest value of ω in Table 2. The third-order translog reduces noise in the data and misspecification issues that can be otherwise captured by the composite error (i.e. $v_{it} = \omega_{it} + u_{it}$) in (2).

Based on the well-known fact that more productive companies selfselect foreign markets, exporters are allowed to operate under a different technology in S9.¹² Controlling for domestic exporters provides on average a 14% higher value of ω_{it} relative to the baseline S3. S9 emphasizes the role of export status in productivity of domestic firms while explaining the importance of accounting for sources of firm heterogeneity in calculating productivity.

5.3. Measuring the effects of FDI knowledge spillovers on TFP

Table 3 shows two stages GMM results from the parameterized law of motion, equation (10). By identifying the specific effects of spillovers on productivity ω_{it} , we are able to compare the results obtained from the present approach with those derived from the standard regression framework (Havránek and Iršová, 2011; Iršová and Havránek, 2013).

Our novel estimation framework indicates that knowledge spillovers in vertical linkages (inter-industry spillovers) are the most critical productivity drivers, whereas horizontal effects from FDI tend to shrink productivity. Compared with a regression framework that measured vertical spillover effects using the same data set Bournakis (2021), the productivity gains from vertical spillovers are much smaller in economic terms. The present values are also smaller than the findings gathered in a meta-analysis study of vertical spillovers (Havránek and Iršová, 2011) that corrects for publication bias (the corresponding $\hat{\theta}_{vertical}$ was revealed in the region of 0.25). Meta-analysis estimates that do not correct for publication bias are even higher.

5.4. An overview of the TFP estimates with endogenous FDI spillovers

The main message from the findings of the first part of the paper is that positive vertical spillover effects do exist, albeit their economic size is not as large as reported in the previous literature. More likely, this is due to the endogeneity bias that underlies the estimates, which assume that FDI spillovers evolve exogenously from the firm's productivity. Intuitively, TFP gains for domestic firms arise from MNEs in downstream and upstream industries rather than from agglomeration effects within the same industry (Lu et al., 2017). Accordingly, when MNEs are customers of domestic firms (backward linkages), the latter provide inputs tailored to the foreign investor's technical and quality requirements. In contrast, when domestic firms purchase inputs from MNEs (forward linkages), these inputs are infused with advanced tacit knowledge that enhances the technical expertise of the buyer.

Regarding horizontal spillovers (intra-industry linkages), MNEs tend to increase competition and "steal" market shares from local firms,

¹² See Temouri et al. (2013) and Schank et al. (2010) and the references thereafter for some relatively recent evidence on the export self-selection hypothesis.

Table 3

Estimates of the productivity process, Equation (11).

	S1	S3	S4	S 6
θ_1	0.99	0.99	0.99	0.99
	(0.01)	(0.05)	(0.03)	(0.01)
9 _{spill}		0.018		0.07
.1		(0.009)		(0.019)
vertical			0.14	
			(0.07)	
) _{inta}				0.026
				(0.003)
abs × spill				1.29
				(0.67)
obs	62,946	62,946	62,946	62,583

which inevitably causes efficiency and productivity losses. The latter argument explains why ω under S5 has a lower value relative to S3 and S4. Negative or insignificant horizontal spillovers are frequently documented in the literature (Gorodnichenko et al., 2014). Organizational capital also contributes to productivity since it serves the dual purpose of stimulating TFP and indirectly improving the firm's capacity for facilitating foreign knowledge. The autonomous role of organizational capital in enhancing TFP in S6 is significant but economically smaller relative to vertical spillovers. In any case, TFP gains from FDI spillovers remain conditional to the firm's absorptive capacity. In conclusion, our analysis confirms that domestic firms in transition (Damijan et al., 2013) as well as in technologically frontier economies (i.e. the group of EU countries in the current sample) are urged to invest in their technological capabilities before benefiting from the MNEs' advanced knowledge.

6. FDI and aggregate productivity: revisiting the puzzle

After allowing for endogenous spillovers in the evolution of productivity, we are well suited to provide a more systematic assessment of the reallocation effects of FDI, which is the second significant contribution of the paper. This section estimates whether FDI presence measured as the share of FDI inflows over total industry output (at the two-digit level) is associated with higher on average productivity of purely domestic firms.¹³ The objective of this empirical investigation is to identify whether a higher industry exposure to FDI initiates a reallocation mechanism among domestic firms. The internal gains in productivity from FDI knowledge spillovers documented in the previous section make domestic firms more productive and enable them to gain market share over their less effective competitors. The reallocation of market segments to more productive firms results in less productive firms losing gradually market share and eventually exiting the market, which increases aggregate (sectoral) productivity. Since the methodology presented in section 3 accounts endogenously for FDI knowledge spillovers within companies, any residual positive effect observed from FDI at the industry level is attributable to a reallocation mechanism that rewards domestic firms with a higher TFP with higher production shares in their industries. In other words, the regression framework that is used by the previous literature (Aitken and Harrison, 1999; Blomström and Sjöholm, 1999; Wooster and Diebel, 2010) in evaluating the role of FDI on the local economy mixes within-firm gains from knowledge spillovers with industry reallocation effects. This section separates the two and acknowledges the different sources of FDI gains.

We view this part of the paper as feeding into the traditional FDI literature (De Mello, 1999; Li and Liu, 2005) that investigates the size of inter-industry effects from FDI due to reallocation dynamics that promote aggregate industry productivity. For the identification of reallocation effects, we specify the following regression:

$$\frac{1}{N}\sum_{i\in j}\widetilde{\omega}_{it} = \delta_0 + \delta_1 \frac{FDI_{jt}}{Y_{jt}} + \pi'_{jt}\delta_\pi + (\lambda_j \times \eta_t) + (\varphi_c \times \eta_t) + v_{jt},$$
(16)

where $\widetilde{\omega}_{it}$ is the weighted TFP adjusted for firm *i*'s share to total output in industry *j* (2-digit NACE classification), $\frac{FDI}{\gamma}$ is the share of FDI inflows to total industry output in industry *j*, π is a vector of other industry-specific characteristics that drive productivity and δ are parameters to be estimated. Additionally, a variant specification (16) is formulated with additional lags (up to two) to account for delays in implementing reallocation effects:

$$\frac{1}{N}\sum_{i\in j}\widetilde{\omega}_{il} = \delta_0 + \sum_{s=0}^2 \delta_{1l-s} \frac{FDI_{jl-s}}{Y_{jl-s}} + \pi'_{jl}\delta_\pi + (\lambda_j \times \eta_l) + (\varphi_c \times \eta_l) + v_{jl}.$$
(17)

Specifications (15) and (16) are augmented with industry λ_j , year η_t and country φ_c fixed effects. More preferably, we use the interaction of industry-year and country-year fixed effects to control for unobserved time-variant industry and time-invariant country factors that matter for TFP movements at the industry level. We are not more explicit for the nature of these idiosyncrasies in the present analysis as the main focus of this application is on the estimated parameter of δ_1 that captures the existence of reallocation effects due to higher FDI shares in the industry. The novelty of these specifications is that the dependent variable is the weighted industry average of firm-specific TFPs as these are derived from a production function that accounts for endogenous FDI knowl-edge spillovers.

Table 4 presents results from (16) considering as dependent variable the value of ω_{it} obtained from frameworks S1, S2, and S3. Note, S1 does not control for spillovers, S2 assumes industry homogeneity, and S3 assumes industry heterogeneity in the derivation of ω_{it} . We apply three econometric estimators for each of these specifications: a pooled OLS (POLS) without controlling for cross-sectional heterogeneity, a Least Squared Dummy Variable (LSDV), and an OLS with industry-year and country-year interaction fixed effects. The industry-specific characteristic in (16) and (17) is the degree of concentration in the industry (Herfindahl-Hirschman index) that measures the type of conduct in the market. Two competing scenarios are possible regarding the effect of market concentration on aggregate productivity. As Nickell (1996) highlights, the exercise of monopolistic power indicates a source of slack, which is detrimental to incentives and effort. As a result, high concentration rates signify misallocation of resources and lower levels of productive efficiency. There is sufficient empirical evidence to support the proposition that increases in the degree of competition enhance aggregate productivity (see among others Holmes and Schmitz Jr., 2010; Aghion et al., 2008). On the other hand, high levels of concentration might be an ex-post symptom of a market reallocation process that shifts shares towards the productivity leaders of the industry. In support of this view, Baqaee and Farhi (2020) finds that aggregate productivity depicts the performance of firms within the top quantile of the distribution rather than within-firm improvements across the entire spectrum. Since we have already considered within-firm productivity improvements (that is, spillovers from FDI and organizational capital), the degree of concentration is likely to correlate with the ability of firms to achieve significant scale economies making new entrants unable to compete on cost efficiency (Tsekouras and Daskalopoulou, 2006). Although an in-depth analysis of the productivity concentration relationship is beyond the scope of the present paper, the econometric estimates in the following tables scrutinize empirically the two competing scenarios providing up-to-date evidence regarding this debate.

The coefficients of δ_1 in Table 4 are statistically insignificant in the first three columns. No reallocation effects are found when aggregate TFP is derived from a specification that does not allow for knowledge

¹³ Appendix G shows average values of FDI share, concentration, and leverage for the sample of the 2-digit NACE industries.

Table 4

Industry reallocation effects from FDI:Baseline specifications.

	$\frac{1}{N}\sum_{i\in j}\widetilde{\omega}_{it}$	$\frac{1}{N}\sum_{i \in j} \widetilde{\omega}_{it}$	$\frac{1}{N}\sum_{i\in j}\widetilde{\omega}_{it}$	$\frac{1}{N}\sum_{i \in j} \widetilde{\omega}_{it}$	$\frac{1}{N}\sum_{i\in j}\widetilde{\omega}_{it}$	$\frac{1}{N}\sum_{i\in j}\widetilde{\omega}_{it}$	$\frac{1}{N}\sum_{i\in j}\widetilde{\omega}_{it}$	$\frac{1}{N}\sum_{i\in j}\widetilde{\omega}_{it}$	$\frac{1}{N}\sum_{i \in j} \widetilde{\omega}_{it}$
	S1			S2			S 3		
	\$1.1	\$1.2	S1.3	\$2.1	S2.2	S2.3	\$3.1	\$3.2	S3.3
	POLS	LSDV	OLS	POLS	LSDV	OLS	POLS	LSDV	OLS
δ_0	-0.135	0.104	3.017	-0.171***	-0.195	-0.291	-0.016	-0.084	-0.071
	(0.21)	(0.77)	(3.02)	(0.05)	(0.12)	(0.25)	(0.04)	(0.09)	(0.34)
δ_1	-0.011	-0.021	-0.017	0.018***	0.018***	0.014***	0.006*	0.006	0.007*
	(0.02)	(0.02)	(0.02)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
$\delta_{Concentration}$	0.022	0.018	0.006	0.018***	0.016***	0.012***	0.006**	0.005	0.003
	(0.03)	(0.03)	(0.02)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)	(0.00)
δ_{Lev}			-0.054			0.005			0.001
			(0.06)			(0.00)			(0.01)
δ_{Lev2}			0.002			-0.000			0.000
			(0.00)			(0.00)			(0.00)
Industry FE	No	Yes		No	Yes		No	Yes	
Country FE	No	Yes		No	Yes		No	Yes	
Year FE	No	Yes		No	Yes		No	Yes	
Industry \times Year	No	No	Yes	No	No	Yes	No	No	Yes
Country \times Year	No	No	Yes	No	No	Yes	No	No	Yes
N	1189	1189	1141	1172	1172	1132	1142	1142	1096
adi. R-sq	0.02	0.11	-0.05	0.66	0.79	0.77	0.14	0.30	0.28
Log lik.	-3249	-3168	-2790	-902	-593	-365	-1025	-888	-569
Clusters	102	102	102	100	100	100	98	98	98

Notes: Cluster robust standard errors at the country and industry level are shown in parentheses. S1 specifications derive ω_{it} without allowing for spillovers in the law of motion. S2 specifications derive ω_{it} assuming industry homogeneity in the estimation of the production function. S3 specifications derive ω_{it} under industry heterogeneity and spillovers in the law of motion.

spillovers at the firm level. S2 and S3 indicate reallocation effects that tend to be larger when the estimation framework of ω_{it} controls for industry heterogeneity. Regarding $\delta_{Concentration}$, the current data set shows that the level of concentration in the industry does not always affect productivity but when it does (in all models of S2 and S3.1) the degree of concentration points to the existence of scale economies and efficiency gains that improve productivity, rather than a source of slack. Models S1.3, S2.3, and S3.3 include the linear and quadratic leverage term to account for a non-linear relationship between the industry's exposure to external financial dependence and investment in productivity-enhancing activities. We have not found such a non-linear effect (i.e. $\hat{\delta}_{Lev} = \hat{\delta}_{Lev2} = 0$ in any of the specifications shown in Table 4. Turning to the economic importance of the estimated coefficients, δ_1 ranges from 0.006 to 0.018, which suggests that a 10% increase in the share of FDI inflows of industry j raises- ceteris paribus- the aggregate productivity of the domestic firms in the industry by 6.1–18%. ¹⁴

Table 5 replicates results from (16) using in $\frac{1}{N} \sum_{i \in j} \widetilde{\omega}_{it}$ the value of ω_{it} obtained from the remaining estimation frameworks, S4–S8, of Table 1. We only consider the LSDV estimator with the interactions of industryyear and country-year FEs. Three out of the five specifications provide a statistically significant coefficient for δ_1 with the economic size of this parameter to be slightly higher than the values shown in Table 4. More precisely, the three specifications that generate significant effects for δ_1 are those that allow for vertical spillovers (S4), horizontal spillovers (S5), and absorptive capacity (S6) in the law of motion of productivity ω_{it} . Considering our baseline estimation framework for calculating ω_{it} S6, the value of coefficient δ_1 suggests that a 10% increase in the share of FDI inflows increases industry TFP up to 15%. Table 5 shows that the elasticity of aggregate productivity from FDI driven reallocation effects is even higher under alternative estimation frameworks of ω_{it} , while the degree of concentration in the industry also impacts productivity (in

¹⁴ Given that ω_{it} is expressed in logs, the elasticity is derived as *exp* (0.006 × 10) = 1.061. Accordingly, a 10% increase in FDI share leads to a 6.1% higher TFP.

Table 5

Industry realloca	tion	effects fr	om FE	I: Results	s under	additional	specifications	;
for the derivation	1 of 1	firm TFP.						

	$\frac{1}{N}\sum_{i\in j}\widetilde{\omega}_{it}$				
	S4	S5	S6	S7	S8
δ_0	-0.687	-0.143	-0.265*	0.054	0.026
	(0.57)	(0.13)	(0.15)	(0.17)	(0.05)
δ_1	0.024**	0.024***	0.015***	0.006	-0.004
	(0.01)	(0.01)	(0.00)	(0.01)	(0.01)
$\delta_{Concentration}$	0.025***	0.011	0.017***	0.017**	-0.000
	(0.01)	(0.01)	(0.00)	(0.01)	(0.00)
Industry $ imes$ Year FE	Yes	Yes	Yes	Yes	Yes
Country \times Year FE	Yes	Yes	Yes	Yes	Yes
N	1126	1140	1178	1126	1110
adj. R-sq	0.52	0.61	0.79	0.71	0.48
Log lik.	-1890	-1635	-569	-1484	-932
Clusters	96	98	101	96	93

Notes: Cluster robust standard errors at the country and industry level are shown in parentheses. S4 derives ω_{it} allowing for vertical spillovers in the law of motion. S5 derives ω_{it} allowing for horizontal spillovers in the law of motion. S6 derives ω_{it} allowing for absorptive capacity in the law of motion. S7 derives ω_{it} allowing for a 2nd order Markov process and spillovers in the law of motion. S8 derives ω_{it} from a 3rd order translog production function allowing for aggregate spillovers in the law of motion. All specifications assume industry heterogeneity in the estimation of the production function.

four out of the five specifications), positively enhancing the scenario that low competition is more likely an ex-post feature of a reallocation process that made productive firms dominate the market.

7. Robustness analysis

As a further test of robustness for identifying within-industry reallocation effects, Table 6 shows results from specification (16) using up to two lags (s = 2) of the FDI share $\frac{FDI_{jl-s}}{Y_{jl-s}}$ variable. The reason for using

Table 6

Industry reallocation	effects	from FDI	with	higher	order	lags.
-----------------------	---------	----------	------	--------	-------	-------

	$\frac{1}{N}\sum_{i\in j}\widetilde{\omega}_{it}$				
	S4	S5	S6	S7	S8
δ_0	-0.229	-0.094	-0.080	-0.082	-0.008
	(0.17)	(0.09)	(0.05)	(0.11)	(0.04)
$\delta_1(t)$	0.000	0.019*	0.011***	-0.003	-0.008
	(0.02)	(0.01)	(0.00)	(0.02)	(0.01)
$\delta_1(t-1)$	-0.012	-0.016	0.001	-0.015	-0.008
	(0.02)	(0.01)	(0.00)	(0.02)	(0.01)
$\delta_1(t-2)$	0.033**	0.012	0.006	0.019**	0.011*
	(0.02)	(0.01)	(0.00)	(0.01)	(0.01)
$\delta_{Concentration}$	0.017**	0.008	0.011***	0.015**	-0.000
	(0.01)	(0.01)	(0.00)	(0.01)	(0.00)
Industry $ imes$ Year FE	Yes	Yes	Yes	Yes	Yes
Country $ imes$ Year FE	Yes	Yes	Yes	Yes	Yes
N	927	935	966	923	912
adj. R-sq	0.5829	0.7001	0.7970	0.7546	0.5757
Log lik.	-1284	-962	-164	-1121	-637
Clusters	94	96	100	93	93

Notes: Cluster robust standard errors at the country and industry level are shown in parentheses. S4 derives ω_{it} allowing for vertical spillovers in the law of motion. S5 derives ω_{it} allowing for horizontal spillovers in the law of motion. S6 derives ω_{it} allowing for absorptive capacity in the law of motion. S7 derives ω_{it} allowing for a 2nd order Markov process and spillovers in the law of motion. S8 derives ω_{it} from a 3rd order translog production function allowing for aggregate spillovers in the law of motion. All specifications assume industry heterogeneity in the estimation of the production function.

higher-order lags lay within the hypothesis that productivity gains from reallocation take some time before being realized in full (Monastiriotis and Alegria, 2011). The time-hysteresis hypothesis implies that as new MNEs enter the domestic market, they gain share over domestic firms, thus causing some market stealing effects and a decline in productivity in the short run for domestic firms. In the long run, however, the group of domestic firms capitalizes on knowledge from the presence of MNEs rebounds by gaining market shares over their less competitive counterparts, which eventually generates substantial productivity gains for the aggregate sector. The introduction of higher-order lags of the FDI variable in Table 6 is expected to capture this time hysteresis.

The use of higher-order lags in FDI in Table 6 now provides statistically significant coefficients for t-2 in S7 and S8. At the same time, previously, the estimates of δ_1 for the same specifications were insignificant in Table 5. In S5 and S6, the reallocation effect of FDI takes place relatively quicker, with only the coefficient of δ_{1t} to be significant. For S4, the estimated coefficient in period *t* turns insignificant, while gains from reallocation become evident in period *t*-2. The main message of the estimation results from equation (17) supports the hysteresis hypothesis as positive FDI effects in aggregate (sectoral) productivity are not always instant. Rather, they occur with time delay; nonetheless, with only one of three coefficients in (17) being statistically significant in each specification, reallocation gains from FDI result in only a one-off boost without significant time persistence.

Overall, sections 6 and 7 highlight the statistical and economic importance of FDI for the aggregate productivity of domestic firms. The highest size of this effect is 0.033 for S4, which signifies productivity gains at the order of 33% after a 10% increase in the FDI share. This particular effect takes place more likely with a two years lag.¹⁵ As our analytical framework treats FDI more systematically by distinguishing

between within-firm knowledge and across-firm reallocation effects, we view the present results as a very encouraging piece of evidence for the positive role of FDI in the domestic economy.

8. Conclusions

8.1. Concluding remarks and policy implications

To the best of our knowledge, our paper is the first endeavor that endogenizes the effect of FDI spillovers in the TFP measurement of domestic firms. In light of this novelty, we revisit the debate on the role of FDI in boosting the productivity of the FDI recipient economies under a new lens. Our framework decomposes the potential FDI effects into withinfirm knowledge spillovers and across-firms industry reallocation effects. In the former, spillovers are assumed to evolve endogenously in the production decisions of domestic firms. At the same time, in the latter, we relate the tendency of more productive firms to capitalize on learning gains and increase market shares. Using an array of different specifications, we explore various industrial linkages through which MNEs affect the performance of their domestic counterparts. The main advantage of endogenizing FDI spillovers is that it corrects for omitted variable bias in the evolution of productivity by accounting specifically for factors that were previously considered to be pure statistical noise leading to spurious estimates of the production inputs. Allowing for endogenous FDI spillovers, TFP estimates are higher, which implies that misleading and inaccurate estimations drive previous results on FDI spillovers. Gains from reallocation are also crucial as increases in FDI inflows by 10% can cause proportionally or even higher increases in aggregate productivity. When firms increase their productivity from FDI spillovers, they trigger a market reallocation mechanism that leads to a higher level of productivity in the aggregate industry. The combination of these effects not distinguishable in the previous literature points to multiple benefits from FDI, which is pretty encouraging for the welfare of the host economy.

Although the role of global value chains in improving productive efficiency is well attributed in the literature, what remains less acknowledged is their role as a conduit of knowledge and technology transfer. In this respect, the approach introduced in our paper provides a better methodological device. It should be regarded as the most appropriate modeling set-up for future research that examines how business interactions in a globalized environment affect the firm's performance.

The main policy lessons from our analysis can be summarized as follows: first, as inter-industry rather than intra-industry linkages are the primary sources of knowledge spillovers, the policy focus should be gathered on the origins of learning from partnerships between MNEs and domestic firms, whether through the sale or purchase of inputs. In this regard, economic policy initiatives should not be restricted to attracting foreign firms, since agglomeration gains from foreign firms (intra-industry) are either insignificant or negative, but rather incentives that encourage MNEs to partner with domestic firms and become embedded in the local economy. The second policy-related message that our work conveys is the substantial firm heterogeneity in how individual firms capitalize on productivity gains from foreign knowledge. This heterogeneity is partly based on the amount of organizational capital within a firm, determining how much the firm can absorb. Knowledge spillovers from FDI are dependent on the firm's internal capability to facilitate external knowledge. Consequently, encouraging firms to upgrade their technological expertise should remain a high priority on the policymakers' agenda.

8.2. Research limitations and future research

The present study develops some new methodological paths in modeling spillovers, but we are far from arguing that the current formulation includes all the possible sources of knowledge spillovers. We only

¹⁵ The methodological derivation of our results differs from the standard approaches used in the FDI industry level literature, nonetheless, even a crude comparison with the previous findings will be useful. Present estimates are higher than productivity gains found in Haskel et al. (2007) for UK domestic firms that do not exceed 0.5%, while contradict the negative FDI effects found in Liu and Wang (2003), Aitken and Harrison (1999), Konings (2001) and Kugler (2006).

consider spillovers derived from MNEs, but our work can be extended to include in the law of motion of productivity factors such as trade from outsourcing materials and services. Trade in materials and services embody tacit knowledge that reflects the R&D effort of the trading partners. Therefore, trade can also serve as a conduit for the international transmission of knowledge. Similarly, domestic inter-industry input transactions can also facilitate R&D spillovers, as has been previously shown by (Bournakis et al., 2018; Giovannetti and Piga, 2017). There is also a regional and geographical element in the presence of spillovers. This is to say that productivity in a firm might be affected by the performance of its peers that are located within a close geographical distance (Iyoha, 2021). These aspects should be addressed in future research, given the availability of suitable data.

Another issue that merits further investigation following our novel methodological framework in this paper is the relationship between concentration and productivity. The current findings contradict the stylized fact that highly concentrated industries suffer from slack and low incentives, which cause productivity degradation. However, our approach already incorporates the firm's internal efforts to improve productivity through foreign learning, which allows the more productive rivals of the industry to become price-competitive and naturally acquire a higher market share. After this reallocation process, the market equilibrium established does not necessarily imply a negative nexus of concentration -productivity. This proposition requires further empirical scrutiny, and it is also another path for future research.

From a more technical perspective, our approach ensures that the estimated production inputs are consistent while performance improvements of the firm are properly identified. An unavoidable limitation is that we perform this, at least, for the most part, using parametric methods. It would be interesting to examine, in future research, which components of our estimation procedure can be robustified further by using non-parametric or, at least, more flexible techniques.

Appendices.

Appendix A. The Evolution of Intangibles 2001–2014



Appendix B. Further Specifications in the derivation of ω_{it}

Spec	Description	mean
S1	CD without spillovers in ω_{it}	4.30
S2	CD with spillovers in ω_{it}	4.29
S3	Translog with spillovers in ω_{it} and domestic MNE as a production input	4.10

Appendix C. Summary Statistics, EFIGE Dataset, 6 European Countries, 2001–14

stats	OPRE	MATE	STAF	FIAS	EMPL
N	79,259	79,259	79,259	79,259	79,259
mean	13,791	7740	2394	5044	62
sd	131,036	100,429	19,317	67,085	322
min	0	0	0	0	10
p1	288	19	96	13	4
p25	1500	492	418	238	14

stats	OPRE	MATE	STAF	FIAS	EMPL
p50	3023	1266	705	665	23
p75	7004	3415	1389	1916	42
p90	18,369	10,062	3256	5365	92
p99	148,520	87,140	24,893	55,183	623
max	11,900,000	11,000,000	1,881,892	6,342,371	19,586

Notes: Values are expressed in Euros 2005 constant prices.

Appendix D. Number of Exporters, MNEs and Domestic Firms

country	Established Exporters	Sporadic Exporters	Foreign Firms
France	352	1173	532
Germany	75	217	178
Hungary	37	109	149
Italy	952	1966	300
Spain	541	2102	300
UK	59	114	291
Total	2016	5681	1750

Notes: Sporadic exporters are domestic firms that export only in 2008, established exporters are domestic firms that export all years during the period 2001–2014. Foreign firms have at least one shareholder of foreign origin that holds at least 10% of capital shares.

Appendix E.	TFP (2001–2014)	of 7699 EU F	irms Under	Different	Structural Sp	ecifications

country	S1	S2	S3	S4	S 5	S6	S7	S8	S9	
FRA	3.494	4.534	3.342	3.787	3.603	4.553	4.856	0.874	4.373	
GER	3.082	4.832	2.159	4.614	2.660	4.331	4.683	0.575	3.392	
HUN	0.787	3.289	0.688	2.942	2.196	3.142	1.898	0.180	2.352	
ITA	3.641	4.435	1.767	3.882	3.680	4.550	6.887	3.914	4.443	
SPA	2.567	3.920	2.861	3.787	2.918	4.022	5.333	0.080	3.498	
UK	2.992	3.986	1.644	3.465	4.571	3.734	6.608	2.622	5.748	

Appendix E presents mean values of ω by country. As expected, substantial differences in ω arise in the cross-sectional dimension of our panel. The ranking of countries based on productivity does not maintain the same pattern across specifications. Under **S3** France, Spain, and Germany have the leadership (Appendix E), whilst under **S4** the ability of German firms to gain from vertical spillovers plays an important role in the evolution of productivity. When it comes to the importance of horizontal spillovers (S5) UK firms benefit the most from the competition pressure induced by MNEs. The same pattern holds when we allow for the export status of the firm.



Appendix F. TFP of Domestic Firms. Estimated from S4

Appendix G. Industry Characteristics in (%)

nace2d	Description	FDI Share	Concentration	Leverage
10	Food products	3.01	22.95	64.40
11	Beverages	19.31	33.38	51.07
12	Tobacco	83.15	76.16	49.61
13	Textiles	6.97	15.56	55.71
14	Wearing Apparel	14.79	30.81	57.94
15	Leather and related products	10.41	8.54	56.83
16	Wood products	12.16	21.77	59.49
17	Paper	14.26	28.75	57.31
18	Printing	23.37	17.58	59.86
19	Coke	48.35	68.41	54.06
20	Chemicals	12.48	17.86	54.30
21	Pharmaceuticals	13.81	43.77	46.87
22	Rubber	7.17	15.02	57.51
23	Other non-metallic	7.65	21.24	50.43
24	Basic Metals	9.46	28.90	63.07
25	Fabricated Metals	13.61	20.93	57.39
26	Electronics	10.69	25.61	52.94
27	Electrical Equipment	23.18	23.94	55.81
28	Machinery	4.17	18.59	57.42
29	Motor Vehicles	9.71	27.44	59.69
30	Transport	32.43	43.57	64.53
31	Furniture	9.48	22.76	59.21
32	Other manufacturing	18.73	29.80	51.94
33	Repair and installation	12.14	39.03	50.99
Mean		13.88	26.36	56.56

Notes: FDI share is the ratio of FDI Inflows to Industry Output. Concentration is Herfindahl-Hirschman Index. Leverage is the sum of current and non-current liabilities over total assets.

References

- Ackerberg, Daniel A., Caves, Kevin, Frazer, Garth, 2015. Identification properties of recent production function estimators. Econometrica 83 (6), 2411–2451.
 Aghion, Philippe, Braun, Matias, Fedderke, Johannes, 2008. Competition and
- productivity growth in South Africa. Econ. Transit. 16 (4), 741–768.
- Aitken, Brian J., Harrison, Ann E., 1999. Do domestic firms benefit from direct foreign investment? Evidence from Venezuela. Am. Econ. Rev. 89 (3), 605–618.
- Aldieri, Luigi, Sena, Vania, Vinci, Concetto Paolo, 2018. Domestic R&D spillovers and absorptive capacity: some evidence for US, Europe and Japan. Int. J. Prod. Econ. 198, 38–49.
- Altomonte, Carlo, Aquilante, Tommaso, 2012. The EU-EFIGE/Bruegel-unicredit Dataset. Tech. Rep. Bruegel Working Paper.
- Altomonte, Carlo, et al., 2013. Internationalization and innovation of firms: evidence and policy. Econ. Pol. 28 (76), 663–700.
- Aw, Bee Yan, Roberts, Mark J., Xu, Daniel Yi, 2011. R&D investment, exporting, and productivity dynamics. Am. Econ. Rev. 101 (4), 1312–1344.
- Baqaee, David Rezza, Farhi, Emmanuel, 2020. Productivity and misallocation in general equilibrium. Q. J. Econ. 135 (1), 105–163.
- Blalock, Garrick, Gertler, Paul J., 2009. How firm capabilities affect who benefits from foreign technology. J. Dev. Econ. 90 (2), 192–199.
- Blomström, Magnus, Sjöholm, Fredrik, 1999. Technology transfer and spillovers: does local participation with multinationals matter? Eur. Econ. Rev. 43 (4–6), 915–923.
- Bloom, Nicholas, Sadun, Raffaella, Van Reenen, John, 2010. Recent advances in the empirics of organizational economics. Annu. Rev. Econ. 2 (1), 105–137.
- Bloom, Nicholas, Sadun, Raffaella, Van Reenen, John, 2012. The organization of firms across countries. Q. J. Econ. 127 (4), 1663–1705.
- Bournakis, Ioannis, 2021. Spillovers and productivity: revisiting the puzzle with EU firm level data. Econ. Lett. 201, 109804.
- Bournakis, Ioannis, Christopoulos, Dimitris, Mallick, Sushanta, 2018. Knowledge spillovers and output per worker: an industry-level analysis for OECD countries. Econ. Inq. 56 (2), 1028–1046.
- Bournakis, Ioannis, Mallick, Sushanta, 2021. Do corporate taxes harm economic performance? Explaining distortions in R&D-and export-intensive UK firms. Macroecon. Dyn. 25 (1), 5–27.
- Crass, Dirk, Peters, Bettina, 2014. Intangible assets and firm-level productivity. ZEW Discuss. Pap. 14.
- Damijan, Jože P., et al., 2013. Impact of firm heterogeneity on direct and spillover effects of FDI: Micro-evidence from ten transition countries. J. Comp. Econ. 41 (3), 895–922.
- De Loecker, Jan, 2013. Detecting learning by exporting. Am. Econ. J. Microecon. 5 (3), 1–21.
- De Loecker, Jan, Goldberg, Pinelopi Koujianou, 2014. Firm performance in a global market. Annu. Rev. Econ. 6 (1), 201–227.
- De Loecker, Jan, Warzynski, Frederic, 2012. Markups and firm-level export status. Am. Econ. Rev. 102 (6), 2437–2471.

De Mello, Luiz R., 1999. Foreign direct investment-led growth: evidence from time series

and panel data. Oxf. Econ. Pap. 51 (1), 133–151.

- Decker, Ryan A., et al., 2016. Declining business dynamism: implications for productivity. In: Brookings Institution, Hutchins Center Working Paper.
- Escribano, Alvaro, Fosfuri, Andrea, Tribó, Josep A., 2009. Managing external knowledge flows: the moderating role of absorptive capacity. Res. Pol. 38 (1), 96–105.
- Farinas, José C., Martín-Marcos, Ana, 2007. Exporting and economic performance: firmlevel evidence of Spanish manufacturing. World Econ. 30 (4), 618–646.
- Giovannetti, Emanuele, Piga, Claudio A., 2017. The contrasting effects of active and passive cooperation on innovation and productivity: evidence from British local innovation networks. Int. J. Prod. Econ. 187, 102–112.
- Gorodnichenko, Yuriy, Svejnar, Jan, Terrell, Katherine, 2014. When does FDI have positive spillovers? Evidence from 17 transition market economies. J. Comp. Econ. 42 (4), 954–969.
- Griffith, Rachel, Redding, Stephen, Van Reenen, John, 2003. R&D and absorptive capacity: theory and empirical evidence. Scand. J. Econ. 105 (1), 99–118.
- Haskel, Jonathan E., Pereira, Sonia C., Slaughter, Matthew J., 2007. Does inward foreign direct investment boost the productivity of domestic firms? Rev. Econ. Stat. 89 (3), 482–496.
- Holmes, Thomas J., Schmitz, Jr, James A., 2010. Competition and productivity: a review of evidence. Annu. Rev. Econ. 2 (1), 619–642.
- Hsieh, Chang-Tai, Klenow, Peter J., 2009. Misallocation and manufacturing TFP in China and India. Q. J. Econ. 124 (4), 1403–1448.
- IMF, 2009. Balance of Payments and International Investment Position Manual. International Monetary Fund, Washington, DC.
- Iršová, Zuzana, Havránek, Tomáš, 2013. Determinants of horizontal spillovers from FDI: evidence from a large meta-analysis. World Dev. 42, 1–15.
- Iyoha, Ebehireme Marie-Terese, 2021. Essays on the Role of Networks in Firm Productivity and International Trade (PhD thesis).
- Javorcik, Beata, 2004. Does foreign direct investment increase the productivity of domestic firms? In search of spillovers through backward linkages. Am. Econ. Rev. 94 (3), 605–627.
- Konings, Jozef, 2001. The effects of foreign direct investment on domestic firms: evidence from firm-level panel data in emerging economies. Econ. Transit. 9 (3), 619–633.
- Kugler, Maurice, 2006. Spillovers from foreign direct investment: within or between industries? J. Dev. Econ. 80 (2), 444–477.
- Levinsohn, James, Petrin, Amil, 2003. Estimating production functions using inputs to control for unobservables. Rev. Econ. Stud. 70 (2), 317–341.
- Li, Xiaoying, Liu, Xiaming, 2005. Foreign direct investment and economic growth: an increasingly endogenous relationship. World Dev. 33 (3), 393–407.
- Liu, Xiaohui, Wang, Chenggang, 2003. Does foreign direct investment facilitate technological progress?: evidence from Chinese industries. Res. Pol. 32 (6), 945–953.
- López, Ricardo A., 2009. Do firms increase productivity in order to become exporters? Oxf. Bull. Econ. Stat. 71 (5), 621–642.
- Lu, Yi, Tao, Zhigang, Zhu, Lianming, 2017. Identifying FDI spillovers. J. Int. Econ. 107, 75–90.
- Marrocu, Emanuela, Paci, Raffaele, Pontis, Marco, 2012. Intangible capital and firms' productivity. Ind. Corp. Change 21 (2), 377–402.

I. Bournakis and M. Tsionas

Melitz, Marc, Levinsohn, James A., 2006. Productivity in a differentiated products market equilibrium. In: Manuscript. Princeton University. Meyer, Klaus E., Sinani, Evis, 2009. When and where does foreign direct investment

Meyer, Klaus E., Sinani, Evis, 2009. When and where does foreign direct investment generate positive spillovers? A meta-analysis. J. Int. Bus. Stud. 40 (7), 1075–1094.

- Monastiriotis, Vassilis, Alegria, Rodrigo, 2011. Origin of FDI and intra-industry domestic spillovers: the case of Greek and European FDI in Bulgaria. Rev. Dev. Econ. 15 (2), 326–339.
- Newman, Carol, et al., 2015. Technology transfers, foreign investment and productivity spillovers. Eur. Econ. Rev. 76, 168–187.
- Nickell, Stephen J., 1996. Competition and corporate performance. J. Polit. Econ. 104 (4), 724–746.
- OECD, 2012a. OECD Structural Analysis, Input-Output Database.
- OECD, STAN, 2012b. OECD Structural Analysis Statistics, vol. 29. Organisation for Economic Co-operation and Development. http://www.oecd-ilibrary.org/industryand-services/data/stan-input-output stan-in-out-data-en. Olley, G. Steven, Pakes, Ariel, 1996. The dynamics of productivity in the
- telecommunications equipment industry. Econometrica 64 (6), 1263–1297.
- Orlando, Michael J., 2004. Measuring spillovers from industrial R&D: on the importance of geographic and technological proximity. In: RAND Journal of Economics. pp.

777–786.

- Rizov, Marian, Walsh, Patrick Paul, 2009. Productivity and trade orientation in UK manufacturing. Oxf. Bull. Econ. Stat. 71 (6), 821–849.
- Schank, Thorsten, Schnabel, Claus, Wagner, Joachim, 2010. Higher wages in exporting firms: self-selection, export effect, or both? First evidence from linked employeremployee data. Rev. World Econ. 146 (2), 303–322.
- Syverson, Chad, 2004. Product substitutability and productivity dispersion. Rev. Econ. Stat. 86 (2), 534–550.
- Temouri, Yama, Vogel, Alexander, Wagner, Joachim, 2013. Self-selection into export markets by business services firms–Evidence from France, Germany and the United Kingdom. Struct. Change Econ. Dynam. 25, 146–158.
- Tsekouras, Kostas D., Daskalopoulou, Irene F., 2006. Market concentration and multifaceted productive efficiency. J. Prod. Anal. 25 (1), 79–91.
 Wooster, Rossitza B., Diebel, David S., 2010. Productivity spillovers from foreign direct
- Wooster, Rossitza B., Diebel, David S., 2010. Productivity spillovers from foreign direct investment in developing countries: a meta-regression analysis. Rev. Dev. Econ. 14 (3), 640–655.